Author’s response to reviews

Title: Contributors to suicidality in rural communities: Beyond the effects of depression

Authors:

Tonelle E Handley (tonelle.handley@newcastle.edu.au)
Kerry J Inder (kerry.inder@newcastle.edu.au)
Frances J Kay-Lambkin (f.kaylambkin@unsw.edu.au)
Helen J Stain (helen.stain@newcastle.edu.au)
Michael N Fitzgerald (michael.fitzgerald@newcastle.edu.au)
Terry J Lewin (terry.lewin@hnehealth.nsw.gov.au)
John R Attia (john.attia@newcastle.edu.au)
Brian J Kelly (brian.kelly@newcastle.edu.au)

Version: 3 Date: 12 November 2011

Author’s response to reviews:

12th November 2011

Dear editorial board of BMC Psychiatry,

Please accept our apologies for the missing information noted in our manuscript “Contributors to suicidality in rural communities: Beyond the effects of depression.” A revised version of this manuscript is attached.

This manuscript now contains a sub-section under Methods, titled Consent and ethical approval. This section contains the requested information, including the procedures by which consent was obtained, and the bodies that gave ethical approval for this study.

Thank you for considering our manuscript.

Tonelle Handley on behalf of the authors